\begin{table}%tD.1
  \caption {\label{tab_sOri_GCS:NM}Photometry and proper motions of known 
sources non selected as potential member candidates by the GCS\@.}
%\centerline
 {  \begin{tabular}{c c c c c c c c c l}
 \hline\hline
RA$^{a}$       & Dec.$^{a}$        &  $Z$   &   $Y$  &   $J$  &   $H$  &  $K$   & $\mu_{\alpha}\cos{\delta}$ & $\mu_{\delta}$ & Comments$^{b}$ \\
 h:m:s           &   d:':''          &  mag   &   mag  &   mag  &   mag  &  mag   &    mas/yr                  &  mas/yr        &                \\
\hline
05:38:38.89 & $-$02:28:01.7 & 16.332 & 15.844 & 15.276 & 14.735 & 14.405 &  $-$0.4 & $-$10.3 & M487350 ([SE 2004]70; blue $Z-J$) ) \\
05:38:41.46 & $-$02:35:52.3 & 15.478 & 99.999 & 13.963 & 13.399 & 12.958 &  $-$0.5 &  $-$0.8 & M500279; No $Y$ photometry \\
05:38:47.90 & $-$02:37:19.4 & 13.148 & 12.771 & 12.205 & 11.449 & 10.908 &  $-$28.8 &  $-$26.3 & M92149 (PM NM) \\
05:38:51.00 & $-$02:49:13.9 & 16.149 & 15.620 & 15.044 & 14.424 & 14.106 &  $-$7.6 &   13.1 & M799173 (blue $Z-J$)  \\
05:39:06.97 & $-$02:12:16.9 & 14.358 & 14.132 & 13.666 & 13.102 & 12.967 &    3.2 &   $-$3.4 & M1462013 (blue $Z-J$)  \\
05:39:07.60 & $-$02:29:05.6 & 16.201 & 15.553 & 14.874 & 14.299 & 13.928 &   43.3 &  $-$7.0 &  M537040 (SOri~20; PM NM) \\
%05:39:25.61 & $-$02:38:43.6 & 16.580 & 16.072 & 15.355 & 14.214 & 13.441 &    5.2 &   $-$3.3 & M633105 (HH446; blue $Z-J$) \\
05:39:26.47 & $-$02:26:15.5 & 14.255 & 13.927 & 13.375 & 12.719 & 12.459 &  $-$0.8 &    0.3 & M856047 (blue $Z-J$) \\
05:40:25.80 & $-$02:48:55.3 & 16.285 & 15.974 & 15.455 & 14.435 & 13.334 &    0.5 &    3.1 & M1701117 (blue $Z-J$) \\
\hline
05:39:16.58 & $-$02:38:25.4 & 20.270 & 19.837 & 19.113 & 18.204 & 18.151 &  --     &  --     &  SOri~26 (M4.5) \\
05:37:07.21 & $-$02:32:44.2 & 16.649 & 16.098 & 15.498 & 14.937 & 14.614 &  7.5    & 12.8    &  SOri~34     \\
05:39:38.50 & $-$02:31:13.3 & 17.870 & 17.285 & 16.590 & 16.061 & 15.719 &  --     &  --     &  SOri~41     \\
05:38:13.96 & $-$02:35:01.3 & 18.335 & 17.591 & 16.926 & 16.376 & 15.966 &  --     &  --     &  SOri~43     \\
05:38:07.11 & $-$02:43:20.9 & 18.602 & 17.882 & 17.172 & 16.667 & 16.244 &  --     &  --     &  SOri~44 (M7)  \\
05:38:23.02 & $-$02:37:55.1 & 99.999 & 19.670 & 18.849 & 18.031 & 17.862 &  --     &  --     &  SOri~49 (M7.5) \\
05:39:47.05 & $-$02:25:24.5 & 99.999 & 19.988 & 18.934 & 99.999 & 17.932 &  --     &  --     &  SOri~57      \\
05:39:10.02 & $-$02:28:11.5 & 15.504 & 15.044 & 14.533 & 13.990 & 13.706 &  20.6   & 23.6    &  SOri J053909.9-022814 (M5) \\
 \hline
05:40:34.61 & $-$02:33:13.8 & 15.103 & 14.675 & 14.115 & 13.489 & 13.204 & $-$6.6   &   12.5  & KJN29, phot NM      \\
05:39:10.02 & $-$02:28:11.5 & 15.504 & 15.044 & 14.533 & 13.990 & 13.706 &   20.6   &   23.6  & KJN33,SOriJ053909.9$-$022814, phot NM \\
05:38:18.17 & $-$02:43:34.9 & 15.473 & 15.137 & 14.636 & 14.036 & 13.775 &    6.6   &    7.6  & KJN34, phot NM      \\
05:40:17.06 & $-$02:26:49.0 & 15.885 & 15.514 & 14.963 & 14.413 & 14.104 &   19.1   & $-$5.7  & KJN45, phot NM      \\
05:38:46.59 & $-$02:19:40.4 & 16.356 & 15.914 & 15.352 & 14.795 & 14.479 &    7.3   &   10.6  & KJN53, phot NM      \\
%05:39:07.60 & $-$02:29:05.6 & 16.201 & 15.553 & 14.874 & 14.299 & 13.928 &   43.3   & $-$7.0  & KJN52,SOri20, PM NM \\
05:39:43.39 & $-$02:53:23.0 & 16.439 & 15.899 & 15.293 & 14.791 & 14.452 &   12.4   & $-$10.7 & KJN59, phot NM      \\
 \hline
05:38:33.34 & $-$02:20:59.4 & 19.968 & 18.905 & 18.026 & 17.244 & 16.598 &  --   &  --   &  SOri~54 (M9.5) \\
05:39:03.60 & $-$02:25:36.6 & 20.469 & 19.062 & 18.362 & 17.900 & 17.067 &  --   &  --   &  SOri~58 (L0)  \\
05:39:26.77 & $-$02:26:14.3 & 17.633 & 16.888 & 16.224 & 15.688 & 15.261 &  --   &  --   &  SOri J053926.8-022614   \\
05:39:48.26 & $-$02:29:14.3 & 17.881 & 17.079 & 16.382 & 15.834 & 15.366 &  --   &  --   &  SOri J053948.1-022914 (M7) \\
 \hline
05:37:18.69 & $-$02:40:22.0 & 13.103 & 12.807 & 12.314 & 12.055 & 11.800 & $-$17.4 &  $-$25.7 & SO~9    \\
05:37:23.05 & $-$02:32:46.7 & 15.126 & 99.999 & 14.159 & 13.592 & 13.319 &  $-$22.3 & $-$14.2 & SO~27   \\
\ldots{}    & \ldots{}     & \ldots{} & \ldots{} & \ldots{} & \ldots{} & \ldots{}  & \ldots{}  & \ldots{} & \ldots{} \\
05:39:56.83 & $-$02:53:14.6 & 20.433 & 19.160 & 18.202 & 17.746 & 17.183 &     --   &     --   &  \\
05:39:57.62 & $-$02:47:36.0 & 13.995 & 13.668 & 13.165 & 12.537 & 12.326 &  $-$7.3 &  $-$4.7 & SO~1296 \\
 \hline
 \end{tabular}
 }
\par
 \smallskip
$^{a}$ Coordinates in J2000 from UKIDSS GCS DR4.
$^{b}$ The last column provides the original names found in the literature.
Note: the first part of the table lists objects from the Mayrit catalogue; the second part lists objects published by \citet{bejar99}, \citet{zapatero00}, and \citet{bejar01}, the third part lists candidates classified as ``Maybe'' by \citet{kenyon05}, the fourth part lists objects lying at the borderline between members and non-members, the last section gives mid-infrared sources published by \citet{hernandez07} and \citet{luhman08c}.
\end{table}